Author’s response to reviews

Title: In favour of the definition "adolescents with idiopathic scoliosis": juvenile and adolescent idiopathic scoliosis braced after ten years of age, do not show different end results. SOSORT Award Winner 2014.

Authors:

Sabrina Donzelli (sabrina.donzelli@isico.it)
Fabio Zaina (fabio.zaina@isico.it)
Monia Lusini (monia.lusini@isico.it)
Salvatore Minnella (salvatore.minnella@isico.it)
Stefano Negrini (stefano.negrini@med.unibs.it)

Version: 4 Date: 22 June 2014

Author’s response to reviews:

Thank you very much for all reviewer useful comments, all the authors really appreciated it, I confirm you that in the JIS group were included patients with a radiographic diagnosis of Juvenile scoliosis never treated before 10 years of age. The first evaluation in our Institute was after the age of 10, but those patients have priorly done x-rays which confirmed the diagnosis of Juvenile idiopathic scoliosis. I specified this better in the paper and highlighted in yellow. Thank you!

p1 VDR1 - was changed in Vitamine D Receptor as suggested by one of the reviewer, the correction is highlighted in yellow, thank you!

If you are submitting a revised version you will need to provide the latest covering letter below.